Author's response to reviews

Title: Doctor-patient interaction in Finnish primary health care as perceived by first year medical students

Authors:

Juhani Miettola (juhani.miettola@uku.fi)
Pekka Mantyselka (pekka.mantyselka@uku.fi)
Tuula Vaskilampi (tuula.vaskilampi@uku.fi)

Version: 3 Date: 1 September 2005

Author's response to reviews:

We have made all the requested formatting changes: We have checked title, anstract, author list, affiliation, section headings, manuscript sections, tables and figures etc. Tables have been incorporated into the main manuscript. The Figure 1 is as a separate document.